Author's response to reviews

Title: Painful vertical diplopia as a presentation of a pituitary mass

Authors:

Shveta Bansal (shveta.ophth@gmail.com)
Kaveri Mandal (kmandal@doctors.org.uk)
Ahmed Kamal (ahmed.kamal@aintree.nhs.uk)

Version: 4 Date: 13 March 2007

Author's response to reviews:

The following further revisions have been made
1. amendments to authors titles and qualifications. Placed after the title
2. authors’ contribution. Placed before acknowledgements section
3. competing interests title added
4.